Author's response to reviews

Title: Ketoacidosis due to low-carb diet in a non-diabetic lactating female: A case report.

Authors:

Magnus Ekelund (magnus.ekelund@med.lu.se)
Louise von Geijer (louise.vongeijer@gmail.com)

Version: 2 Date: 13 February 2015

Author's response to reviews: see over
Cover letter

Dear editor of Journal of Medical Case Reports,

It is with great pleasure that we hereby submit the manuscript titled: "Ketoacidosis due to low-carb diet in a non-diabetic lactating female: A case report"

This paper is about a case of ketoacidosis caused by a combination of low carb, high fat diet and lactation. To the best of our knowledge this would be the first reported case of this potentially life-threatening complication. Since different kinds of diets seem to become increasingly popular, we think that it is very important to report this case.

We think that this case would be of interest for a broad range of physicians such as specialists in internal medicine, endocrinology and obstetrics as well as general practitioners.

The patient has given her written informed consent.

This submission does not contain any figures, tables or any other separate files.

The corresponding author submits this manuscript on behalf of all authors.

In this version we have included a Word file of the CARE Guidelines. However, we think that a flow diagram is non-applicable in this case. All the co-authors’ email addresses have been included in the Title page. The description of the patient’s gender has been changed to ‘female’. Keywords have been included. Finally a list of abbreviations has been included.

Yours Sincerely

Magnus Ekelund
MD, PhD

magnus.ekelund@med.lu.se